JAMA Oncology | Original Investigation

Atezolizumab in Combination With Carboplatin and Survival Outcomes in Patients With Metastatic Triple-Negative Breast Cancer The TBCRC 043 Phase 2 Randomized Clinical Trial

Brian D. Lehmann, PhD; Vandana G. Abramson, MD; E. Claire Dees, MD; Payal D. Shah, MD; Tarah J. Ballinger, MD; Claudine Isaacs, MD; Cesar A. Santa-Maria, MD; Hanbing An, PhD; Paula I. Gonzalez-Ericsson, MD; Melinda E. Sanders, MD; Kimberly C. Newsom; Richard G. Abramson, MD; Quanhu Sheng, PhD; Chih-Yuan Hsu, PhD; Yu Shyr, PhD; Antonio C. Wolff, MD; Jennifer A. Pietenpol, PhD

IMPORTANCE Agents targeting programmed death ligand 1 (PD-L1) have demonstrated efficacy in triple-negative breast cancer (TNBC) when combined with chemotherapy and are now the standard of care in patients with PD-L1-positive metastatic disease. In contrast to microtubule-targeting agents, the effect of combining platinum compounds with programmed cell death 1 (PD-1)/PD-L1 immunotherapy has not been extensively determined.

OBJECTIVE To evaluate the efficacy of atezolizumab with carboplatin in patients with metastatic TNBC.

DESIGN, SETTING, AND PARTICIPANTS This phase 2 randomized clinical trial was conducted in 6 centers from August 2017 to June 2021.

INTERVENTIONS Patients with metastatic TNBC were randomized to receive carboplatin area under the curve (AUC) 6 alone or with atezolizumab, 1200 mg, every 3 weeks until disease progression or unacceptable toxic effects with a 3-year duration of follow-up.

MAIN OUTCOME AND MEASURES The primary end point was investigator-assessed progression-free survival (PFS). Secondary end points included overall response rate (ORR), clinical benefit rate (CBR), and overall survival (OS). Other objectives included correlation of response with tumor PD-L1 levels, tumor-infiltrating lymphocytes (TILs), tumor DNA- and RNA-sequenced biomarkers, TNBC subtyping, and multiplex analyses of immune markers.

RESULTS All 106 patients with metastatic TNBC who were enrolled were female with a mean (range) age of 55 (27-79) years, of which 12 (19%) identified as African American/Black, 1 (1%) as Asian, 73 (69%) as White, and 11 (10%) as unknown. Patients were randomized and received either carboplatin (n = 50) or carboplatin and atezolizumab (n = 56). The combination improved PFS (hazard ratio [HR], 0.66; 95% CI, 0.44-1.01; P = .05) from a median of 2.2 to 4.1 months, increased ORR from 8.0% (95% CI, 3.2%-18.8%) to 30.4% (95% CI, 19.9%-43.3%), increased CBR at 6 months from 18.0% (95% CI, 9.8%-30.1%) to 37.5% (95% CI, 26.0%-50.6%), and improved OS (HR, 0.60; 95% CI, 0.37-0.96; P = .03) from a median of 8.6 to 12.6 months. Subgroup analysis showed PD-L1-positive tumors did not benefit more from adding atezolizumab (HR, 0.62; 95% CI, 0.23-1.65; P = .35). Patients with high TILs (HR, 0.12; 95% CI, 0.30-0.50), high mutation burden (HR, 0.50; 95% CI, 0.23-1.06), and prior chemotherapy (HR, 0.59; 95% CI, 0.36-0.95) received greater benefit on the combination. Patients with obesity and patients with more than 125 mg/dL on-treatment blood glucose levels were associated with better PFS (HR, 0.35; 95% CI, 0.10-1.80) on the combination. TNBC subtypes benefited from adding atezolizumab, except the luminal androgen receptor subtype.

CONCLUSIONS AND RELEVANCE In this randomized clinical trial, the addition of atezolizumab to carboplatin significantly improved survival of patients with metastatic TNBC regardless of PD-L1 status. Further, lower risk of disease progression was associated with increased TILs, higher mutation burden, obesity, and uncontrolled blood glucose levels.

TRIAL REGISTRATION Clinical Trials.gov Identifier: NCT03206203

JAMA Oncol. doi:10.1001/jamaoncol.2023.5424 Published online December 14, 2023.



Author Affiliations: Author affiliations are listed at the end of this article.

Corresponding Author: Jennifer Pietenpol, PhD, Vanderbilt University Medical Center, 652 Preston Research Building, Nashville, TN 37232 (j.pietenpol@vumc.org).

riple-negative breast cancer (TNBC) is a heterogeneous disease characterized by increased genomic instability, immune infiltration,¹ and high programmed death ligand 1 (PD-L1) expression.² These features provided the rationale for early clinical studies evaluating the efficacy of anti-PD-L1 monoclonal antibody atezolizumab as monotherapy or in combination with nab-paclitaxel in advanced TNBC.^{3,4} The durable responses, particularly in PD-L1-positive tumors, led to the phase 3 IMpassion130 trial and accelerated US Food and Drug Administration (FDA) approval for PD-L1-positive metastatic TNBC.³ A subsequent phase 3 trial evaluated atezolizumab in combination with paclitaxel; however, this combination did not improve progression-free survival (PFS) or overall survival (OS) vs paclitaxel alone in PD-L1-positive metastatic TNBC.⁵ The reasons for this are unclear and suggest that anti-PD-L1 therapy efficacy may differ when combined with different chemotherapy. Anti-PD-L1 therapy was first approved in combination with taxane chemotherapy.⁶ Unlike taxanes, platinum agents are DNA intercalating agents and generate neoantigens that may stimulate an immune response.⁷ Therefore, the study reported herein, TBCRC 043 was designed to prospectively evaluate the efficacy of adding the anti-PD-L1 antibody at zolizumab to carboplatin therapy.

TNBC is a heterogeneous disease with varying immune cell composition and may have differential sensitivity to immune checkpoint therapies.^{8,9} Although several biomarkers, such as tumor mutation burden (TMB)^{10,11} and tumor-infiltrating lymphocytes (TILs),¹² are associated with better response, there is a need to identify patients who are unlikely to respond to immunotherapy and spare them from severe immunerelated toxic effects.¹³ We thus present clinical trial results as well as molecular correlates of response to atezolizumab in combination with carboplatin.

Methods

Study Design and Participants

TBCRC 043 is a prospective, multicenter, randomized, doubleblind phase 2 trial (NCT03206203) investigating the efficacy of atezolizumab in combination with carboplatin vs carboplatin alone in patients with metastatic TNBC. Eligible patients had either clinical stage IV or metastatic invasive breast cancer negative for estrogen receptor (<10%), progesterone receptor (<10%), and *ERBB2* (immunohistochemical analysis, \geq 0/1 or fluorescence in situ hybridization, <2.0). Patients with an Eastern Cooperative Oncology Group performance status of 0 to 1, measurable disease, adequate hematologic, kidney, hepatic, and cardiac function, 0 to 1 prior treatments for metastatic disease, and no prior carboplatin in the metastatic setting or prior immune-oncology treatment were eligible. Patients were not stratified by PD-L1 status.

The protocol (Supplement 1) was approved by ethical and institutional review boards (IRB#160633) at the participating institutions, and all patients provided written informed consent and did not receive financial compensation. Patients were screened and enrolled at participating centers of the

Key Points

Question Is the efficacy of carboplatin increased with atezolizumab in metastatic triple-negative breast cancer (TNBC), and are there clinical or molecular correlates associated with response?

Findings In this randomized clinical trial including 106 patients, combining atezolizumab with carboplatin significantly improved progression-free and overall survival of patients with metastatic TNBC; tumor-infiltrating lymphocytes, mutation burden, obesity, and increased blood glucose levels were associated with response to anti-programmed cell death ligand 1 therapy.

Meaning Adding atezolizumab to carboplatin provided a clinically meaningful benefit to patients with metastatic TNBC.

Translational Breast Cancer Research Consortium (TBCRC). Data were collected and reviewed by the Clinical Trials Office and Data and Safety Monitoring Committee.

Treatments and End Points

Patients were randomized to receive intravenous carboplatin area under the curve (AUC) 6 alone or in combination with atezolizumab, 1200 mg, every 3 weeks until disease progression, unacceptable toxic effects, or withdrawal of consent. On disease progression (clinically/Response Evaluation Criteria in Solid Tumors), patients on the carboplatin-alone arm were allowed to cross over to receive atezolizumab alone after undergoing a metastatic biopsy if reasonably safe (eFigure 1 in Supplement 2).

The primary end point was PFS, the time from the first day of treatment to disease progression or death. Secondary end points were overall response rate (ORR; CR + PR) and clinical benefit rate (CBR; CR + PR + stable disease \geq 6 months), duration of response (DOR), and OS. Adverse events were graded using the National Cancer Institute Common Terminology Criteria for Adverse Events, V5.0.

Statistical Analysis

The primary PFS end point was powered for a 1-sided test of the hazard ratio (HR) for PFS at α = .10. The Kaplan-Meier method was used to estimate time-to-event curves and medians for PFS, OS, and DOR. Stratified log-rank tests were used to determine significance. Cox proportional hazards regression models were used to estimate HRs and 95% CIs. Data cutoff was October 2021. Additional details of the statistical analysis are given in the eMethods in Supplement 2. The analysis was performed using R statistical software (version 4.3.0, R Project for Statistical Computing) and followed the Consolidated Standards of Reporting Trials (CONSORT) reporting guidelines.

Results

Study Design and Participants

From August 8, 2017, through October 6, 2020, 130 patients with metastatic TNBC were enrolled on study from 6 participating sites (eFigure 1 in Supplement 2). Twenty-four pa

tients were excluded, and 106 were randomized to receive atezolizumab plus carboplatin (n = 56) or carboplatin (n = 50) (**Figure 1**). Five patients receiving carboplatin and 2 on the combination arm were removed from the study for reasons other than disease progression.

All study participants' sex was female, and demographic characteristics were well-balanced between treatment arms (eTable 1 in Supplement 2). The mean (range) age of participants was 55 (27-79) years, 12 of whom identified as African American/Black (19%), 1 as Asian (1%), 73 as White (69%), and 11 as unknown (10%). PD-L1 positive (>1%) rates were similar between the atezolizumab plus carboplatin (10 [18%]) and carboplatin (10 [20%]) arms. Most (87 [82%]) patients received prior chemotherapy, with 35 patients receiving adjuvant alone (33%), both neoadjuvant and adjuvant (22 [20%]), or meta-static treatment (34 [32%]). Of those receiving metastatic treatment, 12 patients (35%) received treatment as first line. Of the patients who had received prior chemotherapy, 15 (14%) had received a prior platinum agent.

Safety and Tolerability

The median (range) duration of treatment for patients receiving atezolizumab plus carboplatin was 17.4 (1.4-90.3) weeks, whereas for single-agent carboplatin it was 15.4 (3.0-72.1) weeks. The combination was generally well tolerated, and toxic effects were consistent with previous reports.^{5,6} The most common adverse events (AEs, >1) on the combination arm were thrombocytopenia, anemia, lymphocytopenia, nausea, fatigue, and increased liver enzymes (eTable 2 in Supplement 2). Compared with carboplatin, atezolizumab plus carboplatin was associated with a higher incidence of grade 3/4 serious AEs (41% vs 8%). Grade 3 immune-related AEs occurred in 10 patients (5%) receiving the combination, and among possible immune-related AEs were individual cases of colitis, hypothyroidism, retinopathy, and infective myositis (eTable 2 in Supplement 2). Even with increased serious AEs, the withdrawal of study drugs was similar between arms, with 3 patients (6%) receiving carboplatin and 2 patients (4%) receiving the combination coming off study for toxic effects. Crossover patients receiving atezolizumab monotherapy had lower serious AEs (21%) than the combination arm.

Efficacy

The data cutoff for PFS was October 2021. At this date, 94 patients (88.7%) experienced disease progression or death and the median (range) duration of follow-up was 8.9 (0.9-33.2) and 10.3 (0.9-30.0) months for carboplatin and the combination, respectively. Adding atezolizumab to carboplatin significantly (HR, 0.66; 95% CI, 0.44-1.01; log-rank P = .05) improved PFS (**Figure 2**A). Median PFS was 4.1 (95% CI, 2.4-7.0) months with atezolizumab plus carboplatin compared with 2.2 (95% CI, 2.0-4.4) months with carboplatin (eTable 3 in Supplement 2).

Overall, 106 patients were evaluable for CBR at 6 months and ORR with 7 treatment failures due to toxic effects/ nonadherance (Figure 1). The secondary end point of ORR was increased from 8.0% (95% CI, 3.2%-18.8%) for patients on the carboplatin arm to 30.4% (95% CI, 19.9%-43.3%) for the com-

jamaoncology.com



CBR indicates clinical benefit rate; CNS, central nervous system; ORR, overall response rate; OS, overall survival; PFS, progression-free survival.

bination (eTable 3 in Supplement 2). The CBR was 18.0% (95% CI, 9.8%-30.1%) for carboplatin and 37.5% (95% CI, 26.0%-50.6%) for patients receiving the combination (eTable 1 in Supplement 2). Among responding patients, the median duration of response was similar between arms (carboplatin, 14.8 months; 95% CI, 8.2-Inf; combination, 11.6 months; 95% CI, 7.6-17.7) (eTable 3 and eFigure 2B in Supplement 2).

At the cutoff date, the addition of atezolizumab to carboplatin was associated with significantly improved OS in the intent-to-treat population (HR, 0.60; 95% CI, 0.37-0.96; logrank P = .03) from 8.6 to 12.6 months (Figure 2B). Furthermore, when crossover patients were left out of the analysis, OS improved more significantly (HR, 0.46; 95% CI, 0.27-0.79; logrank P = .004) from 7.0 (95% CI, 3.2-10.1) months to 12.6 (95% CI, 10.9-19.9) months (supporting data in eFigure 2C in Supplement 2). Twenty patients who had disease progression while receiving atezolizumab monotherapy (eFigure 1 in Supplement 2). OS was not significantly different (log-rank P = .62) between those patients who received the combination and those who received both monotherapies sequentially (Supporting data in eFigure 2D in Supplement 2).

Clinical Correlates

To better understand clinical variables associated with response, we estimated HRs between the arms using PFS. The overall HR favored carboplatin plus atezolizumab (HR, 0.66;

E4 JAMA Oncology Published online December 14, 2023

Research Original Investigation

P = .05) vs carboplatin alone (**Figure 3**). Women between the ages of 41-64 years (HR, 0.49; P = .009), postmenopausal women (HR, 0.50; P = .01), patients with liver metastases (HR, 0.41; P = .02), and patients who received any prior chemotherapy (HR, 0.59; P = .03) significantly benefited from adding atezolizumab.

An independent reference lab performed PD-L1 IHC (Ventana SP142), and 22.2% (20 of 90) of specimens were positive (>1%) for PD-L1 (eTable 4 in Supplement 3). Of these, most (90%) had PD-L1 positivity only in stromal/immune cells, with only 2 specimens with PD-L1-positive tumor cells. PD-L1 expression differed by metastatic site, with liver (P = .003), bone (P = .005), and brain (P = .05) significantly lower than other sites (supporting data in eFigure 3A in Supplement 2). There was no significant (P = .80) difference between the PD-L1 staining and treatment arm (supporting data in eFigure 3B in Supplement 2). Patients with PD-L1-positive tumors did not receive greater benefit on either arm (eFigure 3C-3E in Supplement 2). However, adding atezolizumab was associated with significantly improved PFS and OS in patients with PD-L1negative tumors (eFigure 3F and 3G in Supplement 2).

TILs have been correlated with immune checkpoint response, and therefore we quantified both intratumor TILs (iTILs) and stromal TILs (sTILs) in pretreatment tumor biopsies.¹⁴ Both iTILs and sTILs varied by metastatic biopsy site, with lung, liver, and bone having lower percentages than other sites (eFigure 4A and 4B in Supplement 2). Tumor biopsies were stratified into high (>5%) and low (<5%) iTILs (eFigure 4C and 4D in Supplement 2). Both high baseline iTILs (HR, 0.12; P = .004) or sTILs (HR, 0.45; P = .007) were associated with a lower rate of progression when receiving the combination compared with carboplatin alone (Figure 3; eFigure 4E-4H in Supplement 2).

Because obesity and diabetes are linked to systemic inflammation,¹⁵ we calculated individual patients' body mass index (BMI, calculated as weight in kilograms divided by height in meters squared) and mean on-treatment blood glucose values. More than 60% of patients had overweight (BMI, 25-30; 29 [27.6%]) or obesity (BMI >30; 30 [34.2%]), with the remaining in the normal range (BMI <25; 40 [38.1%]) (eFigure 5A in

Supplement 2). While receiving therapy, more than 60% of patients had average blood glucose levels within prediabetic (100-125 mg/dL; 49.0%) and diabetic levels (>125 mg/dL; 15.0%), whereas approximately one-third had normal levels (<100 mg/dL) (eFigure 5A in Supplement 2). These levels likely reflect uncontrolled glucose levels because 50% of the patients in the diabetic range and no patients in the normal range were taking antidiabetic medicines. There were no significant differences in mean (SD) BMI (27.4 [7.5] vs 29.3 [7.3]; *P* = .33) or mean (SD) blood glucose levels (114.5 [37.4] vs 109.8 [20.7] mg/dL; P = .39) between treatment arms (eFigure 5B and 5C in Supplement 2). However, there was a trend toward greater benefit from the combination for patients with obesity (HR, 0.52; P = .10) and patients with uncontrolled blood glucose levels at prediabetic (HR, 0.62; P = .13) and diabetic (HR, 0.35; P = .09) levels (Figure 3; eFigure 5D and 5E in Supplement 2). Because blood glucose levels varied through treatment, a timedependent Cox regression analysis was performed and showed glucose levels were associated with worse PFS and OS for patients receiving carboplatin (eFigure 5F and 5G in Supplement 2).

Molecular Correlates

This trial was designed for extensive exploratory biomarker analysis to identify patient subgroups with response to atezolizumab that included: (1) tumor and immune PD-L1 expression, (2) nonsynonymous mutation burden rate, (3) TNBC molecular subtype, and (4) lymphocytes composition by multichannel immunofluorescence (eMethods in Supplement 2).

To identify differentially expressed transcripts and evaluate TNBC molecular subtypes associated with response, we performed RNA sequencing on pretreatment tumor biopsies. After adjusting for metastatic tissue site and sample type (fresh frozen vs formalin-fixed paraffin-embedded), we identified 82 differentially expressed transcripts between patients responding (partial response or stable disease) to carboplatin plus atezolizumab and those with progressive disease (eFigure 6A-6C, eTable 5 in **Supplement** 3). Gene ontology analysis of transcripts elevated in responders showed enrichment in insulin-like growth factor (IGF) transport and uptake, and lipid



Figure 2. Kaplan-Meier Estimate of Progression-Free Survival (PFS) and Overall Survival (OS)

Atezolizumab in Combination With Carboplatin In Metastatic TNBC

Figure 3. Subgroup Analysis of Progression-Free Survival (PFS) in Patients Treated With Carboplatin or Carboplatin Plus Atezolizumab

| | | | Favors Favors | |
|--|-----|-------------------|--|--------|
| Source | No. | HR (95% CI) | atezolizumab + carboplatin carboplatin | P valu |
| All | 106 | 0.66 (0.44-1.01) | | .05 |
| Age, y | | | | |
| ≤40 | 13 | 2.28 (0.47-11.05) | | .30 |
| 41-64 | 67 | 0.49 (0.29-0.83) | | .009 |
| >64 | 27 | 0.86 (0.37-1.98) | | .73 |
| Menopausal status | | | | |
| Post | 69 | 0.50 (0.30-0.85) | | .01 |
| Pre | 32 | 0.95 (0.45-2.03) | | .90 |
| Metastatic site | | | | |
| Lymph node | 54 | 0.64 (0.35-1.15) | | .14 |
| Lung | 50 | 0.60 (0.32-1.08) | | .09 |
| Bone | 36 | 0.90 (0.43-1.86) | | .78 |
| Liver | 35 | 0.41 (0.19-0.89) | | .02 |
| Soft tissue | 9 | 1.02 (0.22-4.62) | | .90 |
| Prior chemotherapy | | | | |
| Yes | 84 | 0.59 (0.36-0.95) | | .03 |
| No | 22 | 0.75 (0.30-1.89) | | .55 |
| Prior platinum chemotherapy | | . , | | |
| Yes | 15 | 0.37 (0.07-1.75) | | .21 |
| No | 91 | 0.71 (0.46-0.12) | | .15 |
| PD-L1 IHC | | . , | | |
| Positive | 20 | 0.62 (0.23-1.65) | | .35 |
| Negative | 70 | 0.56 (0.33-0.94) | | .03 |
| Tumor infiltrating lymphocytes | | | | |
| High | 19 | 0.12 (0.03-0.50) | | .004 |
| Low | 71 | 0.82 (0.50-1.34) | | .44 |
| Stromal lymphocytes | | . , | | |
| High | 57 | 0.45 (0.25-0.80) | | .007 |
| Low | 33 | 0.84 (0.40-1.77) | | .65 |
| BMI | | . , | | |
| Normal | 60 | 0.71 (0.37-1.39) | | .33 |
| Overweight | 29 | 0.92 (0.42-2.01) | | .85 |
| Obese | 36 | 0.52 (0.24-1.15) | | .11 |
| Blood glucose | | | | |
| Normal range (<100 mg/dL) | 37 | 1.00 (0.50-1.96) | | >.99 |
| Prediabetic (100-125 mg/dL) | | 0.62 (0.34-1.14) | | .12 |
| Diabetic range (>125 mg/dL) | 16 | 0.35 (0.10-1.18) | | .09 |
| (120,113,120,110,110,110,110,110,110,110,110,110 | 10 | | 015 0.062 0.250 1 2 | 4 8 16 |
| | | 0.0 | HR (95% CI) | 4 0 10 |

Squares indicate PFS hazard ratios (HRs) from univariate Cox regression models, and 95% CIs are indicated by crossing horizontal lines. The dashed line is the reference HR (0.66) for all patients comparing the 2 treatments. BMI indicates body mass index (calculated as weight in kilograms divided by height in meters squared); PD-L1, programmed death ligand 1.

pathways (eTable 5 in Supplement 3). IGF2, a growth factor capable of binding the insulin receptor and implicated in diabetes,¹⁶ was among the highest transcripts expressed in responding patient tumors (eFigure 6D in Supplement 2).

TNBC subtyping was performed on RNA sequencing from 102 pretreatment biopsies, which resulted in 25% basal-like 1 (BL1), 26% mesenchymal (M), 33% basal-like 2 (BL2), and 15% luminal androgen receptor (LAR) (eFigure 6E in Supplement 2). This distribution in metastatic disease differs from primary TNBC in the Cancer Genome Atlas (35% BL1, 30% M, 20% BL2, and 15% LAR) and likely reflects the reduction of the chemosensitive BL1 subtype from prior chemotherapy treatments (eFigure 6F in Supplement 2).^{17,18} All subtypes tended to benefit more with the addition of atezolizumab, except for the LAR subtype (**Figure 4**; eFigure 6G in Supplement 2).

Because variants generate neoantigens, we performed whole genome DNA sequencing on pretreatment tumor tis-

jamaoncology.com

sues to identify somatic variants. The median nonsynonymous mutation rate ranged from 9.1-371.0 mutations/ megabase (MB) with a median (SD) of 15.7 (40.1) mut/Mb (eFigure 7A in Supplement 2). There was no association between TMB and PFS for patients receiving carboplatin (log-rank P = .63); however, patients with high TMB (>25 mut/ Mb) who received atezolizumab and carboplatin had better PFS (log-rank P = .05) (supporting data in eFigure 7B and 7C in Supplement 2).

As with previous studies, TNBC tumors displayed relatively few recurrent mutations (eFigure 7D in Supplement 2).^{19,20} *TP53* was the most frequently (68.8%) mutated gene (eFigure 8A in Supplement 2). Patients with *TP53* mutant tumors tended to have less favorable outcomes to either carboplatin (HR, 1.3; 95% CI, 0.6-2.6) or the combination (HR, 1.6; 95% CI, 0.8-3.2) (Figure 4; eFigure 8B in Supplement 2). *BRCA1* was mutated in 15.1%, and *BRCA2* mutated in 21.3% of tumors

| Variable | No. | HR (95% CI) | | | | 24070 | izumah J | Favors carboplatin | |
|------------------------------------|------|-------------------|-------|-------|-------|--------|------------|-----------------------|-------------|
| TNBC type | INO. | HR (95% CI) | | | | atezoi | 12uiiiaD י | · carboptatin | carboptatii |
| BL1 | 26 | 0.69 (0.29, 1.60) | | | | | | | |
| M | 20 | 0.68 (0.28-1.60) | | | | | | _ | |
| | | 0.56 (0.24-1.30) | | | | | | | |
| BL2 | 34 | 0.68 (0.30-1.42) | | | | | - | | |
| LAR | 15 | 1.02 (0.32-3.23) | | | | | - | | |
| TMB | 71 | 0.67 (0.40.1.00) | | | | | | _ | |
| Low (<25 mut/Mb) | 71 | 0.67 (0.40-1.08) | | | | | | | |
| High (>25 mut/Mb) | 22 | 0.33 (0.11-1.01) | | | | | | | |
| BRCA1/2 | ~ ~ | 0.46 (0.25.0.04) | | | | | | _ | |
| Wild-type | 61 | 0.46 (0.25-0.81) | | | | | | | |
| Variant | 32 | 0.86 (0.41-1.79) | | | | | | | |
| TP53 | 20 | 0.52 (0.22.4.25) | | | | | | _ | |
| Wild-type | 29 | 0.52 (0.22-1.20) | | | | | | - | |
| Variant | 64 | 0.58 (0.33-1.00) | | | | | | | |
| PIK3CA | | | | | | | | _ | |
| Wild-type | 81 | 0.56 (0.34-0.91) | | | | | | | - |
| Variant | 12 | 0.20 (0.03-1.06) | | _ | | | - | | |
| CD8 | | | | | | | | | |
| High (10.5%-61.7%) | 15 | 0.26 (0.05-1.38) | | | | | - | | |
| Low (0%-9.6%) | 77 | 0.69 (0.43-1.12) | | | | | | | |
| CD8/GZMB | | | | | | | | | |
| High (15.8-175.2/mm ²) | 21 | 0.28 (0.09-0.89) | | | | | - | | |
| Low (<0-14.8/mm ²) | 52 | 0.52 (0.29-0.94) | | | | | | | - |
| CD4 | | | | | | | | | |
| High (10.5-61.7/mm ²) | 15 | 0.70 (0.19-2.64) | | | | | | - | |
| Low (0-9.6/mm ²) | 75 | 0.54 (0.33-0.90) | | | | | | _ | - |
| CD4/FOXP3 | | | | | | | | | |
| High (16.4-485/mm ²) | 15 | 0.13 (0.02-0.71) | | | | - | | | |
| Low (0-14.0/mm ²) | 68 | 0.55 (0.32-0.93) | | | | | - | - | - |
| PD-L1 (CPS >10) | | | | | | | | | |
| Yes | 14 | 1.05 (0.30-3.69) | | | | | - | | - |
| No | 76 | 0.56 | | | | | | | |
| PD-L1 (CPS >1) | | | | | | | | | |
| Yes | 32 | 0.82 (0.37-1.84) | | | | | | | |
| No | 58 | 0.55 (0.33-0.94) | | | | | | | - |
| HLA-A | | | | | | | | | |
| High (82.6%-100%) | 69 | 0.61 (0.36-1.02) | | | | | | | |
| Low (4.7%-79.2%) | 20 | 0.65 (0.25-1.66) | | | | | | | |
| | | 0.0 | 015 (| 0.031 | 0.062 | 0.125 | 0.250 | 0.500 | 1 2 |
| | | | | | | н | R (95% 0 | CI) | |

Figure 4. Subgroup Analysis of Progression of Patients Between Treatment Arms

Forest plot shows the hazard ratios (HRs). Between variable cutoffs on each arm stratified by TNBC type, DNA mutation, and multiplex immunofluorescence markers. BL1 indicates basal-like 1; BL2, basal-like 2; M, mesenchymal; LAR, luminal androgen receptor; TMB, tumor mutation burden; CPS, combined positive score.

(eFigure 8C in Supplement 2). Patients with *BRCA1/2*-mutated tumors tended to respond better to carboplatin (HR, 0.86; 95% CI, 0.41-1.79) (Figure 5; eFigure 8D in Supplement 2). Activating *PIK3CA* mutations were identified in 13% of tumors, with most (63.6%) occurring in the LAR subtype, consistent with prior studies (eFigure 8E in Supplement 2).^{21,22} Patients with activating *PIK3CA* variants had significantly worse survival to either carboplatin alone (HR, 2.96; *P* = .04) or the combination (HR, 2.65; *P* = .03) (Figure 4; eFigure 8F in Supplement 2).

Because TILs were associated with response to atezolizumab plus carboplatin, we developed multiplex-immunofluorescence assays to evaluate tumor-specific (pan-CK), PD-L1, and MHC-I expression, as well as effector (CD8) and helper (CD4) T-cell subsets (eFigure 9A-9D in Supplement 2). While CD8-positive T cells trended toward a worse response regardless of the arm, patients with granzyme B (GZMB) positive activated CD8-positive T cells benefited more from the combination (HR, 0.88; 95% CI, 0.43-1.78) compared with carboplatin alone (HR, 1.46; 95% CI, 0.72-2.93). In contrast, CD4-positive T helper cells were associated with longer PFS on both arms, whereas regulatory T cells (CD4+/FOXP3+) cells were associated with worse response (carboplatin, HR = 3.12; 95% CI, 0.92-11.0; combination, HR = 1.03; 95% CI, 0.51-2.07). Combined positive scoring (CPS) of tumor and immune PD-L1-positive cells was associated with longer PFS on both arms at CPS greater than 1, and to a lesser extent at CPS higher than 10 (Figure 5). Because antigen presentation is necessary for tumor-immune recognition, we evaluated MHC-I expression (HLA-A) in tumor epithelium. Tumor HLA-A expression ranged from 4.7%-100%, with a median (SD) of 97.9% (28.4%) (eFigure 9E in Supplement 2). Regardless of the treatment arm, low tumor-specific HLA-A expression (<80%) was associated with decreased PFS (carboplatin, HR = 1.90; 95% CI, 0.97-3.69; combination, HR = 1.56, 95% CI, 0.81-2.97) (Figure 5).



Forest plot shows the hazard ratio (HR) for the progression of patients between variable cutoffs on each arm stratified by TNBC type, DNA mutation, and multiplex immunofluorescence markers. BL1 indicates basal-like 1; BL2,

basal-like 2; M, mesenchymal; LAR, luminal androgen receptor; TMB, tumor mutation burden; CPS, combined positive score.

Discussion

In this study, adding atezolizumab to carboplatin therapy increased PFS from 2.2 to 4.1 months. This increase is similar to adding pembrolizumab (5.6 to 7.5 months) to a gemcitabine-carboplatin therapy.²³ The combination of atezolizumab and carboplatin also increased OS in IIT patients with metastatic disease from 8.6 to 12.6 months.

Patients receiving atezolizumab monotherapy after disease progression on carboplatin had similar OS with fewer toxic effects than those receiving the combination, suggesting sequential chemotherapy and immunotherapy may be another option for those patients in which clinical toxic effects management is critical. However, additional studies are needed for validation because this study was not powered for that comparison and treatment crossover may inherently select for patients with less aggressive disease.

It is unclear which biomarkers are useful for predicting immune checkpoint therapy response. Atezolizumab was initially approved for PD-L1-positive metastatic TNBC,⁶ and pembrolizumab showed benefit in patients with PD-L1-positive disease.²⁴ However, benefit of atezolizumab added to paclitaxel was not observed in patients with PD-L1-positive disease in the IMpassion131 study.⁵ In this study, PD-L1 status did not affect PFS or OS with immunotherapy. However, the incidence of PD-L1-positive tumors (22%) was substantially lower than the 41%, 45%, and 38% reported in the IMpassion130,⁶ IMpassion131,⁵ and KEYNOTE-355 clinical trials,²⁴ respectively. Lower PD-L1 positivity may be attributed to the proportion of biopsies evaluated from primary vs metastatic sites. In this study, most (75%) were metastatic biopsies compared with 48% in the IMpassion131 trial.⁵ Furthermore, the PD-L1 positivity rates vary by metastatic biopsy location, with liver, skin, and bone metastases displaying much lower rates than other sites.²⁵ Regardless, there was a significant increase in PFS and OS in PD-L1-negative tumors consistent with the KEYNOTE-522 trial, in which pembrolizumab provided a benefit regardless of PD-L1 status.²⁶

Like other studies, we observed increased survival benefit for patients receiving the combination whose tumors had higher pretreatment TILs. Multi-immunofluorescence analyses demonstrated that although CD8 T cells were associated with shorter PFS on both arms, activated (GZMB⁺) T cells were associated with improved PFS only on the combination arm. Furthermore, CD4 T cells were associated with increased PFS, while FOXP3-positive regulatory T cells were associated with decreased PFS for patients receiving the combination.

Mutation burden has been associated with immunotherapy response retrospectively²⁷ and prospectively.²⁶ We observed a similar association in patients with TMB-H tumors, having increased PFS on both arms. Activating *PIK3CA* mutations were the only variant associated with a lack of response to immunotherapy. However, *PIK3CA* variants were also associated with decreased clinical benefit with carboplatin alone, consistent with a prior study²⁸ in which patients with *PIK3CA* variants had a lower response to anthracycline-based neoadjuvant chemotherapy.

In this study, we observed an association of obesity and uncontrolled blood glucose levels with better responses to PD-L1 immunotherapy. Obesity has been associated with increased efficacy of PD-1/PD-L1 blockade and attributed to T-cell dysfunction,²⁹ and retrospective studies have shown that patients with obesity with breast cancer achieved the highest benefit.³⁰ This benefit could be attributed to higher adipose tissue composition in the breast and augmented by metabolic syndrome conditions such as type 2 diabetes. Changes in blood glucose levels in fasting-mimicking diets profoundly affect peripheral blood cell composition by reducing immunosuppressive myeloid and regulatory T-cell compartments.³¹ Patients with with obesity with elevated blood glucose levels may have more immunosuppressive immune cells upon which anti-PD1/PD-L1 therapies can act. Future studies are needed to validate these findings and delineate the effects of blood glucose and obesity.

Limitations

Although preliminary associations were observed in the molecular subgroup analyses, caution should be exercised in interpretation because these were retrospective, exploratory, and limited by small patient numbers and require further investigation in larger cohorts to draw definitive conclusions.

Conclusions

In this multicenter phase 2 randomized clinical trial, adding atezolizumab to carboplatin significantly increased PFS and OS in patients with metastatic TNBC. Increased TILs, higher TMB, obesity, and uncontrolled blood glucose levels were associated with a decreased risk of progression, whereas tumor HLA-A expression, *PIK3CA* variants, and the LAR subtype were associated with a greater risk of progression for patients receiving the combination. Crossover patients receiving sequential chemotherapy and immunotherapy had fewer toxic effects than patients receiving the combination, suggesting sequential treatment could be considered for patients whose treatment-related toxic effects are of concern.

ARTICLE INFORMATION

Accepted for Publication: August 29, 2023. Published Online: December 14, 2023. doi:10.1001/jamaoncol.2023.5424

Open Access: This is an open access article distributed under the terms of the CC-BY License. © 2023 Lehmann BD et al. *JAMA Oncology*.

Author Affiliations: Department of Medicine, Vanderbilt University, Nashville, Tennessee (Lehmann, V. G. Abramson); Vanderbilt-Ingram Cancer Center, Vanderbilt University Medical Center, Tennessee (Lehmann, V. G. Abramson, Gonzalez-Ericsson, Sanders, Newsom, Pietenpol); Department of Medicine and Lineberger Comprehensive Cancer Center, University of North Carolina, Chapel Hill (Dees); Department of Medicine, University of Pennsylvania, Philadelphia (Shah); Department of Medicine, Indiana University, Indianapolis (Ballinger); Department of Medical Oncology, Lombardi Cancer Center, Georgetown University, Washington, DC (Isaacs); Department of Medicine, Johns Hopkins School of Medicine, Baltimore, Maryland (Santa-Maria, Wolff); Department of Otolaryngology, Vanderbilt University Medical Center, Nashville, Tennessee (An); Breast Cancer Research Program, Vanderbilt University Medical Center, Nashville, Tennessee (Gonzalez-Ericsson); Department of Pathology, Microbiology and Immunology, Vanderbilt University, Nashville, Tennessee (Sanders); Department of Biomedical Engineering, Vanderbilt University School of Engineering, Nashville, Tennessee (R. G. Abramson); Department of Biostatistics, Vanderbilt University Medical Center, Nashville, Tennessee (Sheng, Hsu, Shyr); Center for Quantitative Sciences, Vanderbilt University Medical Center, Nashville, Tennessee (Sheng, Hsu, Shyr); Department of Biochemistry, Vanderbilt University, Nashville, Tennessee (Pietenpol).

Author Contributions: Drs Pietenpol and Abramson had full access to all of the data in the study and take responsibility for the integrity of the data and the accuracy of the data analysis. Drs Lehmann and Abramson contributed equally to the work.

Concept and design: Lehmann, V. Abramson, Shyr, Wolff, Pietenpol.

Acquisition, analysis, or interpretation of data: All authors.

Drafting of the manuscript: Lehmann, V. Abramson, Newsom, Hsu, Pietenpol.

Critical review of the manuscript for important intellectual content: Lehmann, V. Abramson, Dees, Shah, Ballinger, Isaacs, Santa-Maria, An, Gonzalez Ericsson, Sanders, R. Abramson, Sheng, Shyr, Wolff, Pietenpol.

Statistical analysis: V. Abramson, Sheng, Hsu, Shyr. Obtained funding: V. Abramson, Pietenpol. Administrative, technical, or material support: V. Abramson, Dees, Ballinger, Santa-Maria, Gonzalez Ericsson, Sanders, Newsom, Pietenpol. Supervision: Lehmann, V. Abramson, Pietenpol.

Conflict of Interest Disclosures: Dr V. Abramson reported grants from Genentech during the conduct of the study. Dr Shah reported Advisory board participation from Daiichi Sankyo, Gilead Sciences, and Biotheranostics outside the submitted work. Dr Ballinger reported personal fees from Medscape and TotalCME outside the submitted work. Dr Isaacs reported grants from TBCRC to institution and personal fees from Genentech during the conduct of the study; personal fees from PUMA, Seattle Genetics, AstraZeneca, Novartis, Pfizer, Sanofi, Gilead, ION, Wolters Kluwer, McGraw Hill, SideOut Foundation, and GSK outside the submitted work. Dr Santa-Maria reported grants from Genentech, Pfizer, Novartis, AstraZeneca, Merck, and BMS outside the submitted work. Dr Shvr reported grants from the National Institutes of Health (NIH)/ National Cancer Institute (NCI) during the conduct of the study. Dr Pietenpol reported grants from NCI CA098131 and CA068485, grants from Susan G. Komen, other from Genentech (sponsored the clinical trial and provided study drug), and other from TBCRC (provided support for the clinical trial during the conduct of the study): in addition. Drs Pietenpol and Lehmann are inventors (US Patent No. 11,788,147) of intellectual property (TNBCtype) licensed by Oncocyte Corp; the licensed IP is indirectly related to the work. No other disclosures were reported

Funding/Support: This work was supported by National Cancer Institute grant CA098131 (to J.A.P.), Susan G. Komen grants SAB210301 (to J.A.P.), CCR13262005 (to B.D.L.), Genentech, and the Translational Breast Cancer Research Consortium. We thank Vanderbilt Technologies for Advanced Genomics (VANTAGE), the Data Sciences Shares Resource, and the Translational Pathology Shared Resource for providing technologies supported by NCI grant CAO68485 and NIH S10 instrumentation grants [RR028106, RR027764 and OD023475].

Role of the Funder/Sponsor: The funding agencies had no role in the design and conduct of the study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; and decision to submit the manuscript for publication.

Disclaimer: Dr Shyr is Associate Editor for Statistics for *JAMA Oncology* but was not involved in any of the decisions regarding review of the manuscript or its acceptance.

Data Sharing Statement: See Supplement 3.

Additional Contributions: We thank all the patients who generously volunteered to participate in this study and the TBCRC investigators, research nurses, and study coordinators for their efforts on behalf of the patients. The authors also thank Susan G. Komen and The Breast Cancer Research Foundation for funding support to the TBCRC.

REFERENCES

 Loi S, Michiels S, Salgado R, et al. Tumor infiltrating lymphocytes are prognostic in triple negative breast cancer and predictive for trastuzumab benefit in early breast cancer: results from the FinHER trial. Ann Oncol. 2014;25(8):1544-1550. doi:10.1093/annonc/mdu112

2. Mittendorf EA, Philips AV, Meric-Bernstam F, et al. PD-L1 expression in triple-negative breast cancer. *Cancer Immunol Res*. 2014;2(4):361-370. doi:10.1158/2326-6066.CIR-13-0127

3. Emens LA, Cruz C, Eder JP, et al. Long-term clinical outcomes and biomarker analyses of atezolizumab therapy for patients with metastatic triple-negative breast cancer: a phase 1 study. *JAMA Oncol.* 2019;5(1):74-82. doi:10.1001/jamaoncol. 2018.4224

4. Adams S, Diamond JR, Hamilton E, et al. Atezolizumab plus nab-paclitaxel in the treatment of metastatic triple-negative breast cancer with 2-year survival follow-up: a phase 1b clinical trial. *JAMA Oncol.* 2019;5(3):334-342. doi:10.1001/ jamaoncol.2018.5152 5. Miles D, Gligorov J, André F, et al; IMpassion131 investigators. Primary results from IMpassion131, a double-blind, placebo-controlled, randomised phase III trial of first-line paclitaxel with or without atezolizumab for unresectable locally advanced/metastatic triple-negative breast cancer. *Ann Oncol.* 2021;32(8):994-1004. doi:10.1016/j. annonc.2021.05.801

6. Schmid P, Adams S, Rugo HS, et al; IMpassion130 Trial Investigators. Atezolizumab and nab-paclitaxel in advanced triple-negative breast cancer. *N Engl J Med.* 2018;379(22):2108-2121. doi:10.1056/ NEJMoa1809615

7. O'Donnell T, Christie EL, Ahuja A, et al. Chemotherapy weakly contributes to predicted neoantigen expression in ovarian cancer. *BMC Cancer*. 2018;18(1):87. doi:10.1186/s12885-017-3825-0

8. Lehmann BD, Colaprico A, Silva TC, et al. Multi-omics analysis identifies therapeutic vulnerabilities in triple-negative breast cancer subtypes. *Nat Commun*. 2021;12(1):6276. doi:10. 1038/s41467-021-26502-6

9. Lehmann BD, Jovanović B, Chen X, et al. Refinement of triple-negative breast cancer molecular subtypes: implications for neoadjuvant chemotherapy selection. *PLoS One*. 2016;11(6): e0157368. doi:10.1371/journal.pone.0157368

10. Thomas A, Routh ED, Pullikuth A, et al. Tumor mutational burden is a determinant of immune-mediated survival in breast cancer. *Oncoimmunology*. 2018;7(10):e1490854. doi:10. 1080/2162402X.2018.1490854

11. Alva AS, Mangat PK, Garrett-Mayer E, et al. Pembrolizumab in patients with metastatic breast cancer with high tumor mutational burden: results from the targeted agent and profiling utilization registry (TAPUR) study. *J Clin Oncol*. 2021;39(22): 2443-2451. doi:10.1200/JCO.20.02923

12. Denkert C, Loibl S, Noske A, et al. Tumor-associated lymphocytes as an independent predictor of response to neoadjuvant chemotherapy in breast cancer. *J Clin Oncol*. 2010; 28(1):105-113. doi:10.1200/JC0.2009.23.7370

13. Johnson DB, Balko JM, Compton ML, et al. Fulminant myocarditis with combination immune checkpoint blockade. *N Engl J Med*. 2016;375(18): 1749-1755. doi:10.1056/NEJMoa1609214 14. Tumeh PC, Harview CL, Yearley JH, et al. PD-1 blockade induces responses by inhibiting adaptive immune resistance. *Nature*. 2014;515(7528):568-571. doi:10.1038/nature13954

15. Rohm TV, Meier DT, Olefsky JM, Donath MY. Inflammation in obesity, diabetes, and related disorders. *Immunity*. 2022;55(1):31-55. doi:10.1016/ j.immuni.2021.12.013

16. Casellas A, Mallol C, Salavert A, et al. Insulin-like growth factor 2 overexpression induces β -cell dysfunction and increases beta-cell susceptibility to damage. J Biol Chem. 2015;290(27):16772-16785. doi:10.1074/jbc.M115.642041

17. Masuda H, Harano K, Miura S, et al. Changes in triple-negative breast cancer molecular subtypes in patients without pathologic complete response after neoadjuvant systemic chemotherapy. *JCO Precis Oncol.* 2022;6:e2000368. doi:10.1200/PO. 20.00368

18. Masuda H, Baggerly KA, Wang Y, et al. Differential response to neoadjuvant chemotherapy among 7 triple-negative breast cancer molecular subtypes. *Clin Cancer Res.* 2013;19(19):5533-5540. doi:10.1158/1078-0432.CCR-13-0799

19. Shah SP, Roth A, Goya R, et al. The clonal and mutational evolution spectrum of primary triple-negative breast cancers. *Nature*. 2012;486 (7403):395-399. doi:10.1038/nature10933

20. Staaf J, Glodzik D, Bosch A, et al. Whole-genome sequencing of triple-negative breast cancers in a population-based clinical study. *Nat Med.* 2019;25(10):1526-1533. doi:10.1038/ s41591-019-0582-4

21. Lehmann BD, Abramson VG, Sanders ME, et al; Translational Breast Cancer Research Consortium. TBCRC 032 IB/II multicenter study: molecular insights to AR antagonist and PI3K inhibitor efficacy in patients with AR⁺ metastatic triple-negative breast cancer. *Clin Cancer Res.* 2020;26(9):2111-2123. doi:10.1158/1078-0432.CCR-19-2170

22. Lehmann BD, Bauer JA, Schafer JM, et al. PIK3CA mutations in androgen receptor-positive triple negative breast cancer confer sensitivity to the combination of PI3K and androgen receptor inhibitors. *Breast Cancer Res.* 2014;16(4):406. doi: 10.1186/s13058-014-0406-x

23. Cortes J, Cescon DW, Rugo HS, et al; KEYNOTE-355 Investigators. Pembrolizumab plus chemotherapy versus placebo plus chemotherapy for previously untreated locally recurrent inoperable or metastatic triple-negative breast cancer (KEYNOTE-355): a randomised, placebo-controlled, double-blind, phase 3 clinical trial. *Lancet.* 2020;396(10265):1817-1828. doi:10. 1016/S0140-6736(20)32531-9

24. Cortes J, Rugo HS, Cescon DW, et al; KEYNOTE-355 Investigators. Pembrolizumab plus chemotherapy in advanced triple-negative breast cancer. *N Engl J Med*. 2022;387(3):217-226. doi:10. 1056/NEJM0a2202809

25. Rozenblit M, Huang R, Danziger N, et al. Comparison of PD-L1 protein expression between primary tumors and metastatic lesions in triple negative breast cancers. *J Immunother Cancer*. 2020;8(2):e001558. doi:10.1136/jitc-2020-001558

26. Schmid P, Cortes J, Pusztai L, et al; KEYNOTE-522 Investigators. Pembrolizumab for early triple-negative breast cancer. *N Engl J Med*. 2020;382(9):810-821. doi:10.1056/NEJMoa1910549

27. Yarchoan M, Hopkins A, Jaffee EM. Tumor mutational burden and response rate to PD-1 inhibition. *N Engl J Med*. 2017;377(25):2500-2501. doi:10.1056/NEJMc1713444

28. Guo S, Loibl S, von Minckwitz G, Darb-Esfahani S, Lederer B, Denkert C. PIK3CA H1047R mutation associated with a lower pathological complete response rate in triple-negative breast cancer patients treated with anthracycline-taxane-based neoadjuvant chemotherapy. *Cancer Res Treat.* 2020;52(3):689-696. doi:10.4143/crt.2019.497

29. Wang Z, Aguilar EG, Luna JI, et al. Paradoxical effects of obesity on T cell function during tumor progression and PD-1 checkpoint blockade. *Nat Med.* 2019;25(1):141-151. doi:10.1038/s41591-018-0221-5

30. Yoo SK, Chowell D, Valero C, Morris LGT, Chan TA. Outcomes among patients with or without obesity and with cancer following treatment with immune checkpoint blockade. *JAMA Netw Open*. 2022;5(2):e220448. doi:10.1001/jamanetworkopen. 2022.0448

31. Vernieri C, Fucà G, Ligorio F, et al. Fasting-mimicking diet is safe and reshapes metabolism and antitumor immunity in patients with cancer. *Cancer Discov*. 2022;12(1):90-107. doi: 10.1158/2159-8290.CD-21-0030